June 7th, 2022

Dear Editor,

Please find enclosed the original article revised and re-titled “Analysis of hepatitis C virus positive organs in liver transplantation”.

We are thankful to the reviewers for their comments and suggestions, which were thoroughly considered. We believe that these comments have been highly constructive and useful and improved our manuscript's quality. The manuscript has been modified as kindly suggested by the reviewers. All the points suggested by the three reviewers have been answered in the document shown below.

We hope that these changes fulfill the requirements to make the manuscript suitable for publication in World Journal of Hepatology.

Sincerely yours,

Isabel Legaz

Department of Legal and Forensic Medicine, Biomedical Research Institute (IMIB), Regional Campus of International Excellence "Campus Mare Nostrum", Faculty of Medicine, University of Murcia, E-30110 (Murcia), Spain. Fax: (+34) 868834307 Phone: (+34) 868883957. Email: isalegaz@um.es
ANSWER TO THE REVIEWERS

Manuscript ID: 77963
Title: Analysis of hepatitis C virus positive organs in liver transplantation

The manuscript has been improved according to the REVIEWERS' suggestions as follows. We have added the answers to their related questions (bold) to facilitate understanding of the replies. All changes are marked in red in the manuscript. The authors appreciate the effort to evaluate to improve our work.

REVIEWER #1
Thank you very much for considering our interesting manuscript. We appreciate the time spent on improving our MS.

REVIEWER #2
We appreciate the comments you make to improve the quality of our work and the time of dedication provided. The manuscript has been improved according to the suggestions of REVIEWER #1.

All changes are marked in red in the manuscript. We hope that these changes fulfill the requirements to make the manuscript suitable for publication in World Journal of Hepatology.

Sincerely yours.